Author's response to reviews

Title: Challenges and Opportunities Associated with Neglected Tropical Disease (NTD) and Water, Sanitation and Hygiene (WASH) Intersectoral Integration Programs

Authors:

Elizabeth A Johnston (eajohnston87@gmail.com)
Jordan Teague (jlteague2011@gmail.com)
Jay P Graham (jgraham@GWU.EDU)

Version: 2 Date: 10 October 2014

Author's response to reviews:

I made the requested revisions to the manuscript, which included:
Adding line numbering and double spacing to the document; and
Moving tables to bottom of the text document and including a legend for each table.